Author's response to reviews

Title: A novel mutation in the tyrosine kinase domain of ERBB2 in hepatocellular carcinoma

Authors:
- Tanios Bekaii-Saab (Tanios.Bekaii-Saab@osumc.edu)
- Nita Williams (Nita.Williams@osumc.edu)
- Christoph Plass (Christoph.Plass@osumc.edu)
- Miguel Villalona Calero (Miguel.Villalona@osumc.edu)
- Charis Eng (engc@ccf.org)

Version: 5 Date: 13 November 2006

Author's response to reviews:

Statements about competing interests and authors contributions were both included in the manuscript again after the acknowledgements and before the reference sections. Sincerely - Tanios